Author's response to reviews

Title: Measuring dementia carers' unmet need for services - an exploratory mixed method study

Authors:

Christine Stirling (Christine.Stirling@utas.edu.au)
Sharon Andrews (Sharon.Andrews@utas.edu.au)
Toby Croft (toby.croft@dhhs.tas.gov.au)
James Vickers (James.Vickers@utas.edu.au)
Paul Turner (paul.turner@utas.edu.au)
Andrew Robinson (Andrew.Robinson@utas.edu.au)

Version: 4 Date: 10 May 2010

Author's response to reviews:

The following revisions have been made:

Author list: the list of authors in the manuscript are written exactly as they are in the submission system, both in style and order.

It is now clear under author contributions that all authors read and approved the final manuscript.

Title: The full stop has been removed from the title.

Tables: the table title is now only one sentence long.

Tables- the visible vertical lines have been removed from the table

The following change has not been made despite attempting to follow online instructions:

Figures
that the figures are cropped as closely as possible to minimise white space around the image.

No other changes have been made.

Regards

Christine Stirling